Author's response to reviews

Title: Excellent outcomes among HIV+ children on ART, but unacceptably high
pre-ART mortality and losses to follow-up: a cohort study from Cambodia

Authors:

Marie-Eve Raguenaud (eve_raguenaud@hotmail.com)
Petros Isaakidis (petrosisaakidis@yahoo.com)
Rony Zachariah (rony.zacharia@brussels.msf.org)
Vantha Te (cdc_takeo@camintel.com)
Seithabot Soeung (varun@angkorhospital.org)
Kazumi Akao (kazumiakao0520@hotmail.com)
Varun Kumar (varun@angkorhospital.org)

Version: 6 Date: 13 August 2009

Author's response to reviews: see over
Cover letter

August 13, 2009
By Marie-Eve Raguenaud

Title of manuscript: Excellent outcomes among HIV+ children on ART, but unacceptably high pre-ART mortality and losses to follow-up: a cohort study from Cambodia.

Reference 3: unchanged as this article has not yet been published
Reference 6: details of the reference provided
Reference 11: full details of website given
Figure title: removed
Previous exhibition: moved to the Acknowledgement section
In addition, the text was revised and corrected for typing errors.